Author’s response to reviews

Title: Risk factors for neck pain among forklift truck operators. A retrospective cohort study.

Authors:
Ulf Flodin (ulf.flodin@regionostergotland.se)
Bo Rolander (bo.rolander@rjl.se)
Håkan Löfgren (hakan.lofgren@rjl.se)
Blerim Krapi (blerim.krapi@regionostergotland.se)
Fredrik Nyqvist (fredrik.nyqvist@regionostergotland.se)
Charlotte Wåhlin (charlotte.wahlin@regionostergotland.se)

Version: 4 Date: 26 Jan 2018

Author’s response to reviews:
Respons to reviewers and editor,

Title: Risk factors for neck pain among forklift truck operators. – A retrospective cohort study

Authors: Flodin U, Rolander B, Löfgren H, Krapi B, Nyqvist F, Wåhlin C

Once again, thank you so much for the input on our manuscript. We have made adjustments according to your suggestions, item 1-12. You will see the changes in this submitted version or our manuscript. However, they are not marked since this is the final version.